COMMENT Open Access



Reflections of patient and public involvement from a commissioned research project evaluating a nationally implemented NHS programme focused on diabetes prevention

Rhiannon E. Hawkes^{1*}, Caroline Sanders², Claudia Soiland-Reyes^{3,4}, Lisa Brunton⁵, Kelly Howells², Sarah Cotterill⁶, Carole Bennett⁷, Eric Lowndes⁷, Manoj Mistry⁷, Helen Wallworth⁷ and Peter Bower⁵

Abstract

Patient and Public Involvement and Engagement (PPIE) in research is recognised by the National Institute for Health and Care Research as crucial for high quality research with practical benefit for patients and carers. Patient and public contributors can provide both personal knowledge and lived experiences which complement the perspectives of the academic research team. Nevertheless, effective PPIE must be tailored to the nature of the research, such as the size and scope of the research, whether it is researcher-led or independently commissioned, and whether the research aims to design an intervention or evaluate it. For example, commissioned research evaluations have potential limits on how PPIE can feed into the design of the research and the intervention. Such constraints may require re-orientation of PPIE input to other functions, such as supporting wider engagement and dissemination. In this commentary, we use the 'Guidance for Reporting Involvement of Patients and the Public' (GRIPP2) short form to share our own experiences of facilitating PPIE for a large, commissioned research project evaluating the National Health Service Diabetes Prevention Programme; a behavioural intervention for adults in England who are at high risk of developing type 2 diabetes. The programme was already widely implemented in routine practice when the research project and PPIE group were established. This commentary provides us with a unique opportunity to reflect on experiences of being part of a PPIE group in the context of a longer-term evaluation of a national programme, where the scope for involvement in the intervention design was more constrained, compared to PPIE within researcher-led intervention programmes. We reflect on PPIE in the design, analysis and dissemination of the research, including lessons learned for future PPIE work in large-scale commissioned evaluations of national programmes. Important considerations for this type of PPIE work include: ensuring the role of public contributors is clarified from the outset, the complexities of facilitating PPIE over longer project timeframes, and providing adequate support to public contributors and facilitators (including training, resources and flexible timelines) to ensure an inclusive and considerate approach. These findings can inform future PPIE plans for stakeholders involved in commissioned research.

*Correspondence: Rhiannon E. Hawkes Rhiannon.Hawkes@manchester.ac.uk Full list of author information is available at the end of the article



Plain English Summary

This commentary uses the 'Guidance for Reporting Involvement of Patients and the Public' checklist to describe the activities and lessons learned from the work of a Patient and Public Involvement and Engagement (PPIE) group working with researchers evaluating the National Health Service (NHS) Diabetes Prevention Programme (a nine-month behavioural support programme offered to adults who are at risk of developing type 2 diabetes). The evaluation aimed to find out if the NHS Diabetes Prevention Programme was meeting its aim of reducing type 2 diabetes whilst offering value for money. The PPIE group was made up of members of the public who were either living with diabetes, at risk of diabetes, or had a family history of diabetes. Their aim was to support the work of the research team to identify research priorities, design the research, clarify the meaning of the research results, and help communicate the results to the wider public. However, given that the programme was an NHS service already being delivered in practice, there were limits to what the PPIE group could do in terms of designing the intervention. Instead, our public contributors were involved in all stages of conducting the research, with greater focus on communication of findings from the research evaluation. This report provides an opportunity to reflect on PPIE in a longer-term evaluation of a national programme and reflect on involvement in dissemination and policy impact. We hope that future evaluation programmes can learn from our experiences and optimise PPIE for similar national evaluations.

Keywords Patient and public involvement, Reflections, Diabetes prevention, Type 2 diabetes, National programmes, Evaluation programmes

Background

Patient and public involvement and engagement

Patient and Public Involvement and Engagement (PPIE) is defined by INVOLVE (now known as the National Institute for Health and Care Research [NIHR] Centre for Engagement and Dissemination) as "research carried out 'with' or 'by' members of the public who are actively involved in the research projects" [1]. Public involvement is now an essential requirement for research funded by NIHR in the United Kingdom (UK) and for funders globally [2–4], to ensure that research is of high quality and of practical benefit [1].

Involving members of the public in research improves both quality and relevance of the research. They provide both personal knowledge and experiences of using a service or living with a health condition, which complements that of the research team [1, 5]. In addition to being adequately supported financially, effective PPIE needs to be tailored to the nature of the research, such as the size and scope of the research, whether the research is researcher-led or independently commissioned by a research funder, and whether the research

aims to design an intervention or evaluate it. For example, independent commissioned research evaluations may have limits on how PPIE can inform the design of the intervention when it is already implemented in routine practice. Thus, ensuring a PPIE strategy is in place is important for all types of health research, including diabetes [6].

We reflect on PPIE work conducted as part of an independent commissioned research evaluation called 'DIPLOMA' (Diabetes Prevention Long-term Multimethod Assessment) evaluating the NHS Diabetes Prevention Programme (see Box 1 for more information about the NHS Diabetes Prevention Programme and Box 2 for more information about DIPLOMA). The DIPLOMA PPIE group is the focus of this commentary, which has been co-authored with members of the group. The PPIE members' experiences are included throughout the article and are reflected using the terms 'we' and 'our'.

Focus of this commentary

This commentary uses the 'Guidance for Reporting Involvement of Patients and the Public' (GRIPP2)

Box 1 The NHS Diabetes Prevention Programme

Type 2 diabetes is a global public health priority [7], and 13.6 million people in the UK are now at an increased risk of developing the condition [8]. Type 2 diabetes is preventable by behavioural modifications such as improved diet, increased physical activity and weight loss. Prevention programmes implemented worldwide provide behaviour change interventions for individuals at risk. Trials and real-world studies have found such programmes to be effective at promoting weight loss and thereby reducing the risk of type 2 diabetes [9].

In response to this, the NHS Diabetes Prevention Programme was launched in 2016; a nine-month behaviour change intervention for adults in England who have been identified as having elevated blood glucose levels and therefore at an increased risk of developing type 2 diabetes [10]. The programme is commissioned by NHS England and provides education and support on how to improve diet and increase physical activity to achieve weight loss. The NHS Diabetes Prevention Programme is the largest in the world to achieve universal national coverage [11].

Box 2 The DIPLOMA research project

The 'DIPLOMA' research project (Diabetes Prevention Long-term Multimethod Assessment) is a commissioned mixed-methods study delivered by a multidisciplinary team, designed to provide a rigorous longer-term assessment of the NHS Diabetes Prevention Programme to ensure it is meeting the aim of reducing type 2 diabetes in England in a way that is sustainable and cost-effective [12]. This programme of research was funded by NIHR in 2017 and consisted of eight work packages to evaluate: access and equity of those who enter the NHS Diabetes Prevention Programme, implementation of the programme across England, service delivery and fidelity of the programme, patient outcomes within the programme, and long-term cost effectiveness. As part of the PPIE strategy, DIPLOMA researchers were keen to involve public contributors from the outset. We wanted to involve those who were living with diabetes, were at risk of diabetes, or had a family history of diabetes, and could provide valuable insights on how to design, implement and disseminate research findings to members of the public in a way that was accessible and engaging.

short form checklist, a tool for reporting of PPIE in health and social care [13], and applies it to PPIE in the DIPLOMA research project. GRIPP2 was developed to enhance the quality and transparency of the evidence base for PPIE, so that readers can learn from PPIE in other studies and understand what the result of this involvement was [13]. The short checklist encourages research teams to report five topics: aims, methods, study results, discussion, and reflections of PPIE in research (Additional File 1). We reflect on what went well and the challenges we faced, so that others can learn from our experiences. In particular, we reflect on the challenges and lessons learned in facilitating PPIE involvement in commissioned research for a longerterm evaluation of a programme that was already implemented as an NHS service.

We also reflect on collaboration in terms of recently published UK standards for effective PPIE [14]. These six standards list a clear set of statements for effective public involvement and encourages behaviours such as flexibility, sharing and learning, and respect for others. Two of the UK standards that are particularly relevant in relation to research evaluations closely connected to national policy are 'communications' and 'impact' [14]. We hope that other researchers, members of the public and research funders can learn from our experiences and optimise public involvement for similar national evaluations.

Public involvement and engagement in the DIPLOMA research project reported using GRIPP2

Aim

The aim of the PPIE involvement group in the DIPLOMA research project was for the research team to work collaboratively with public contributors to identify research priorities, design the research, analyse and interpret the results and disseminate research findings to lay audiences, ensuring that patients and the public were informed about ongoing developments throughout the life of the DIPLOMA evaluation. In addition to the PPIE group, DIPLOMA recruited two lay members to our

Study Steering Committee (SSC) group, whose role was, together with external clinicians and expert researchers, to provide overall supervision for the project on behalf of the NIHR [15]. The work of the SSC lay members is not part of this commentary.

Methods

Pre-funding: identifying research priorities

Prior to funding for DIPLOMA, patients and the public were actively involved in preparing the application for this programme of research. An advert was circulated via an existing PPIE group and via a PPIE network. The research team recruited six people (living with diabetes, at risk, or with family history), varying in age, gender, ethnicity and disabilities. Researchers provided a summary of the research plans and asked what they would like to know about the effectiveness of the NHS Diabetes Prevention Programme, services, organisations and participants. Many responses mapped onto the team's existing questions, including: effect on diabetes prevention and other health outcomes; cost effectiveness; equal access; and understanding implementation and service delivery. New topics were suggested that were added to potential research plans: choice of service, and ability to cope with the risk of diabetes; the impact of wider social networks on uptake; change in GP referrals over time; take-up by people with new diagnosis or established conditions; and clarity of information provided at time of referral.

Post-funding: involvement in the DIPLOMA research project

PPIE in the DIPLOMA project is summarised in Table 1. Five of the six public contributors involved in developing the research priorities pre-funding continued their involvement when DIPLOMA was funded. The involvement of public contributors in the NHS Diabetes Prevention Programme design was somewhat limited given that (a) this was a commissioned research evaluation where funders set out a broad outline to answer specific research questions, and (b) the NHS Diabetes Prevention Programme was already implemented in routine practice. Since the aim of DIPLOMA was to provide recommendations for the wider implementation of the programme

Table 1 Involvement of Public Contributors in the DIPLOMA Research Project

Public contributors

A DIPLOMA PPIE group was formed in 2017; there have been a total of 10 contributors throughout the DIPLOMA evaluation, and five who have regularly contributed since 2020

PPIE members were recruited either from: (a) a previous PPIE group who expressed an interest in the DIPLOMA project due to personal and family experiences, (b) a PPIE network called 'Research for the Future', a collaboration between NIHR Clinical Research Network: Greater Manchester, North West EHealth, Northern Care Alliance NHS Group and Health Innovation Manchester, or (c) Diabetes UK

Contributors included representation from people at risk of type 2 diabetes, people with a family history of diabetes, and people from different backgrounds including ethnic minorities

Two lay members were also recruited to our Study Steering Committee group, alongside clinicians and researchers outside of the research team

Meetings

Members of the research team facilitated the PPIE group throughout the five-year project. Everyone was considered equal The PPIE group met 18 times in total (initially face-to-face, and then virtually during and following the Covid-19 pandemic)

Payments

All PPIE members were paid for their involvement, in line with current guidance [17]

PPIE members were reimbursed for any travel expenses and for their time, including meeting attendance and any feedback activity undertaken throughout the research evaluation, as per the current guidance [17]

PPIE involvement

During these meetings, and via email and telephone communication in between meetings, members helped to design research studies within the different work packages, interpret research findings, and co-produce public engagement materials to disseminate research findings for lay audiences

to improve outcomes for patients, we therefore planned to have an equal focus on public 'engagement' to disseminate the research findings to members of the public throughout the DIPLOMA project, keeping members of the public informed of the findings through blogs, videos and lay summaries. DIPLOMA had therefore budgeted for engagement activities throughout the duration of the research, an example of a best practice activity to support stakeholder engagement [16].

Study results

PPIE contribution was specifically for the DIPLOMA project, rather than for the NHS Diabetes Prevention Programme, as the service had already been developed and was implemented in routine practice at the time of the research evaluation. Therefore, public contributors were involved during all phases of the research, from the design and preparation of studies, conduct and implementation, and dissemination of research findings [15].

Design of research studies

Prior to the start of DIPLOMA, the research team worked with six members of the public in writing the funding bid: they raised new ideas about what research questions we should ask, advised the team about how to involve people in the study, and suggested the important role the public can have in engagement. Once the PPIE group was set up, members provided input on the wording of study materials including, but not limited to: topic guides, consent forms, information sheets and study questionnaires to ensure these were understandable and written in Plain English. Members also helped the researchers identify other areas of exploration to

ask during participant interviews. For example, for one of the research topic guides the group suggested that it would be beneficial to ask the interviewees if they felt their expectations of the NHS Diabetes Prevention Programme had been met, and the group also highlighted that interviewees may need breaks factored into the interview. Researchers have also explored recruitment procedures with the PPIE group; we have provided suggestions for strategies such as approaching charities and places of worship, displaying posters in GP surgeries, and posting on social media platforms and specific online groups. As public contributors, being involved early during the research studies ensured that the research was acceptable to potential participants, and is considered a best practice approach [16].

Data analysis and interpretation

Some of our PPIE group have been involved in early stages of qualitative data analysis, for example, reading through a sample of anonymised transcripts of healthcare professional referrals of patients onto the programme, and providing their views on developing themes. Members of our group have also commented on quantitative results from studies, which can be more difficult to convey to a lay audience. For example, researchers asked the PPIE group on their interpretations of quality-of-life data from the NHS Diabetes Prevention Programme.

Researchers from all DIPLOMA work packages have presented findings to the PPIE group at various stages of the evaluation, and we have provided feedback and further interpretation of the findings. Members have fed back on the terminology used and some of the graphs which were too technical for lay members to understand

initially. The researchers produced alternative presentation formats, which the PPIE group fed back on. This regular dialogue between the research team and PPIE group [16] ensured we were able to input on study results, provide interpretations of results, and suggest ways to present the results and complicated statistics to a wider audience.

The PPIE group have also advised on analysis plans, for example, analysis of uptake to the NHS Diabetes Prevention Programme. This dataset included data fields collected by primary care when patients registered on the programme. Some of our group highlighted that religious reasons for declining the programme and the time of year that the programme was declined (e.g., to account for religious holidays) were also important considerations for programme uptake, which was not captured in the dataset that the DIPLOMA research team had access to. In this case, the PPIE group provided valuable insight about how data collection for uptake could be optimised. However, it was not in the remit of the research team to make these direct changes and we could only feedback to commissioners.

Dissemination of research findings

Our PPIE group were passionate about getting involved in raising awareness about diabetes prevention. For example, we raised ideas about wider community engagement and making information as accessible as possible, especially in communities and networks where there would likely be more risks related to inequalities in diabetes and challenges for addressing this. However, it was out of the remit of the research team to 'promote' the NHS Diabetes Prevention Programme, although the PPIE group were able to provide valuable input for communications and dissemination of DIPLOMA findings.

Over the five-year project, we have collaborated with video producers, illustrators and scriptwriters to create a range of accessible web-based videos to promote public engagement of different aspects of DIPLOMA. This has included three animation videos describing (a) the planned DIPLOMA evaluation at the start of the project [18], (b) the overall research findings at the end of the project [19], and (c) information about making changes to health behaviours [20]. We have also co-produced two 'Talking Heads' videos [21, 22]; the first video summarised a qualitative study on how service users understood their type 2 diabetes risk [23], and the second video summarised research on service user take-up and experiences of the NHS Digital Diabetes Prevention Programme [24]. These have been disseminated in the public domain (e.g., via social media channels and NHS England Health Care Innovation Expo). The PPIE group co-produced the video scripts, inputted into the storyboards, and provided

feedback on the overall 'look and feel' of the videos to ensure they were appealing and understandable for the public. We also featured in some of these videos.

The group have also co-produced lay summaries of studies, working closely with DIPLOMA researchers. This again ensured that our research findings were of wider interest to a non-academic audience. We have worked with an illustrator to produce illustrated summaries to disseminate research findings in different formats for diverse audiences. Disseminating research to the general public was particularly important for DIPLOMA, given that it was taxpayer's money subsidising both the research and the NHS Diabetes Prevention Programme, and patients and the public have a right to influence what is supported [25]. We have provided feedback on how to optimise engagement and understanding of these materials for a lay audience, particularly commenting on use of complex statistics and acronyms. PPIE members have also written blogs about their experiences of involvement in evaluations of national programmes [26, 27].

Discussion and conclusions

PPIE has been vital throughout the life of the DIPLOMA evaluation, from study design through to dissemination of findings. As public contributors, we have provided a unique and broader perspective on the work, which increased the potential of the DIPLOMA research to meet the needs of patients and members of the public.

However, PPIE contribution for a commissioned research evaluation of an NHS service already implemented had unique challenges. For example, there have been several occasions where the PPIE group have provided valuable feedback on the NHS Diabetes Prevention Programme, but it was not within the remit of the DIPLOMA research team to make changes to the programme itself. The purpose of the PPIE group was instead to support a research evaluation of an intervention developed by a third party, rather than supporting a research project designing the intervention; an approach much less direct than what some contributors might have been used to. The literature on public involvement has previously identified that one of the most common issues in PPIE work is unclear definition of roles and expectations of public contributors [25]; this could have been better addressed in DIPLOMA and we further reflect on this below.

Given that there was less scope for our PPIE group to provide direct feedback on the NHS Diabetes Prevention Programme itself, we were keen to give at least equal focus to 'engagement' as well as 'involvement', to produce accessible materials on findings for the public. We view this as a particular strength of our approach as this gave the PPIE a higher profile than might be the case with other research projects. Reflecting on the UK standards for PPIE [14], we were involved in developing a high number of accessible co-produced communications for a high profile policy-related national programme [18–22], demonstrating a clear focus on both 'communications' and 'impact' [14]. However, engagement work also came with its own challenges to ensure we kept a clear distinction between dissemination of the research findings and 'promotion' of the NHS Diabetes Prevention Programme.

Finally, national evaluations of programmes such as the NHS Diabetes Prevention Programme are often of greater complexity in terms of methods and data analyses, as was the case for DIPLOMA. For example, this research involved complex health economics and effectiveness analyses, and some public contributors found it more challenging to get involved in these quantitative components. This highlights an example of why there is a greater need for support, resources and training for both public contributors and group facilitators in large commissioned research projects such as DIPLOMA, and PPIE input on these components can take considerable time for all team members. A previous commentary on PPIE contribution in diabetes research has discussed ways in which they have achieved this, such as yearly training on research methods though interactive quizzes for patients [6]. In relation to the UK PPIE standards [14], there could have been some further learning and development opportunities identified for both PPIE contributors and the research team to help facilitate this challenge in DIPLOMA. However, the researchers very much valued the input from our PPIE group on presenting the research findings in a way that was more understandable for lay audiences, and PPIE feedback was taken on board.

Reflections/critical perspective

Our PPIE group agreed that being involved in DIPLOMA was a positive experience where we felt listened to and our feedback was incorporated throughout the entire project; we discussed our views during a meeting specifically organised to reflect on each aspect of the research evaluation. Being part of this research team gave the opportunity for members to learn about how academic research is conducted. Importantly, it also increased our awareness of the constraints involved in both academic research and the implementation of programmes such as the NHS Diabetes Prevention Programme.

Lessons learned in the transition to online meetings

Partway through the project, the Covid-19 pandemic changed the way we were able to work together; it was no longer possible to meet in-person, thus we decided to use a video conferencing platform to host meetings. Although the group have missed face-to-face meetings,

there have been some advantages to online meetings and this new way of working. For example, travelling to the university campus for in-person meetings was at times daunting for some members, and could take up a substantial amount of their day for a two-hour meeting. Online meetings have been more time efficient and cost-effective. Thus, the group reflected the importance of still meeting up in-person, whilst also embracing this new way of working online. Similar issues of adapting to PPIE during the pandemic have been discussed in more depth elsewhere [28].

However, this new way of working together has taught the research team some lessons which we will learn from. For example, one of our public contributors did not have access to a camera for the online meetings, which made both visual communication and feeling included difficult for this individual. It was correctly highlighted by another member of our PPIE group that such access issues would not arise in a face-to-face environment, as every effort would have been made to help members 'get into the room. Whilst online working was new to all of the research team during the Covid-19 pandemic, it is something we have now all become accustomed to. On reflection, this would have been an issue easily dealt with by the research team to ensure all PPIE members had equality of access and inclusive opportunities; one of the UK standards of PPIE [14]. The option to also receive some training or practice of how to use video conferencing during the transition to online meetings could have been another way to ensure the PPIE group felt included and confident in participating in online meetings and is recognised as a best practice activity for stakeholder involvement [16].

Challenges of being part of a longer-term research project

Our group appreciated timely and effective communication from the research team, even when all communication transferred online. However, DIPLOMA has faced its unique challenges due to the project being a lengthy and complicated programme of research, and much longer in comparison to some research projects. This resulted in a lack of continuity of researchers taking on a facilitator role over such a long programme of work, with three different PPIE research facilitators during the fiveyear project (REH, LB and KH), and a turnover of PPIE contributors, both of which had an impact on continuity and group dynamics. Although this was unavoidable, this had at times resulted in unintended consequences. Firstly, there was a loss of contact with original members of the PPIE group who stepped away from the group due to health issues. Secondly, newer members recruited to the PPIE group later in the project had to quickly understand and deal with the complexities of the different work packages in DIPLOMA. Although the research team provided an introduction to the DIPLOMA research when new members of the PPIE group joined, there was a lot of information for new members to take on board for a large project.

Despite DIPLOMA being a complex project with eight work packages, our group felt that the work packages had been usefully broken up to ease understanding. However, some members reported to not always comprehend where their contribution fitted into the wider project. Although during early PPIE meetings there was discussion of timelines to help show where PPIE would be sought at different time points, these had been somewhat impacted by the switch to working online. As with other policy programmes, the NHS Diabetes Prevention Programme had shifting timelines beyond the control of the research team, which made PPIE work more complex.

It was also noted that at times the duration between the meetings were too long. This led to members forgetting about some of the previous work; a particular difficulty of facilitating PPIE for a longer-term evaluation. To mitigate this, a meeting schedule could have been planned in advance for upcoming months of PPIE meetings to reduce the time between meetings when trying to find suitable dates to accommodate the group. Although the above highlights a general issue in PPIE work about desires to avoid burdening the group compared with neglecting the group [29], a more formal procedure could have been put in place for public contributors to regularly feedback any potential issues to the research team so that they could be detected earlier on during the research evaluation. For example, PPIE contribution in diabetes research has previously reported an anonymous survey to work well in obtaining feedback from the group [6].

Challenges of being part of a national evaluation of a commissioned NHS service

Some of our group reflected on the difficulties of PPIE contribution for a research project where the NHS Diabetes Prevention Programme was already in service, rather than an intervention being developed by the research team. It was at times frustrating for PPIE members when we were not able to have a direct impact on the design and delivery of the NHS Diabetes Prevention Programme. Although views were noted and could be fed back to commissioners, the research team had a mediating role between the PPIE group and programme developers. On reflection, the research team could have better explained at the outset to the group what was (not) in the remit of DIPLOMA to ensure better expectation management for working together [14]. The clarification of roles of public contributors in research is discussed in the literature [25, 29] and described as a best practice activity which should be established early on during the research project [15].

We also reflected on our input on the public engagement resources produced by DIPLOMA, including the blogs and videos. We discussed that DIPLOMA has been successful in informing the public via the PPIE group about the research, which were seen as examples of genuine collaboration. However, it was noted that getting this information out to the general public is a more difficult task, as there is already a lot of information on social media about diabetes and uptake on some of these resources was lower than we had hoped. This was anticipated to be a challenge, given that most people seeking diabetes resources would be looking for help and support with their self-management rather than research on that issue, and thus our expectations on resource uptake were modest.

Considerations prior to the commencement of research

The use of the project name 'DIPLOMA' was confusing for us as public contributors, as the acronym bears no meaning with the NHS Diabetes Prevention Programme. Such acronyms used by academics can hinder understanding among public contributors; for example, when PPIE members tried to find out more information about the DIPLOMA project, internet searches brought up information about education courses rather than the NHS Diabetes Prevention Programme. It was reflected that although project acronyms can ease communication internally within the research team, it may not be an accessible shorthand for public contributors, and the wider public, to understand. Going forward, it was discussed that any acronyms and project names could be agreed at the start of the project with the PPIE group who are an integral part of the team and need to be comfortable with using them.

The PPIE group were paid for their involvement throughout DIPLOMA, in line with NIHR guidance [17]. However, one of our PPIE members experienced difficulties in receiving cash reimbursements as this was considered as income by their local Job Centre office, and hence impacted the benefits they were entitled to receive. Alternative arrangements were therefore put in place by the PPIE facilitator to provide payment to this member using vouchers instead. This, however, took considerable time to put in place and the PPIE member received a late payment as a result. Researchers should consider ensuring that they include arrangements in their PPIE strategy for the option to pay public contributors in vouchers, prior to receiving project funding.

Table 2 Recommendations for PPIE in commissioned research evaluations of nationally implemented programmes

Pre-funding of the research project

Researchers could work with public contributors to agree on a project name that the whole group feel comfortable using, avoiding the use of complicated acronyms where possible

Consider alternative remuneration methods for PPIE members (e.g., voucher payments), based on their preferences or potential adverse impact on their income/benefit entitlements

Start of the research project

Provide detailed information about the project timelines, roles of the PPIE group, and scope of the PPIE input, to ensure all members are informed at the start of the process on what is expected

Be explicit about the type of research that public contributors are getting involved in, for example, distinguishing between research that designs a service and research that evaluates a service

Gather ideas from the PPIE group at the start of the project on how they feel they could contribute to the research to ensure that the purpose of public involvement has been jointly agreed

Throughout the research project

Group facilitators could plan a schedule for upcoming meetings in advance, to ensure meetings accommodate for everyone's preferences (including those of the researchers and PPIE members) and to avoid having long breaks in between meetings

If there are long breaks between meetings, PPIE members should be kept up to date with developments from the project

The involvement of someone who is independent of the research team could collate formal feedback from the PPIE group on a regular basis to provide contributors with more opportunity and scope to feedback on any issues that need addressing throughout the duration of the research

Longer project timelines may inevitably result in a turnover of PPIÉ research facilitators and/or public contributors. Measures should therefore be put in place to ensure handover between group facilitators, inductions for new members, and minimal disruption for the group if this occurs

The research team should ensure adequate support is provided to PPIE members, in training and resources, as well as in time, recognising their role as voluntary contributors who have their own commitments outside of the research project

Resources to consider

If using remote involvement, facilitators should provide training in how to use video conferencing to ensure all members feel at ease with the technology before meetings. Nonetheless, face-to-face meetings also remain important to allow the development of relationships and trust within the group, especially at the start of their involvement when the group are new to their PPIE role. Thus, if using remote involvement, a hybrid approach with the opportunity for both in-person and online meetings would be beneficial

Researchers should consider the extra resources and training that may be required for (a) remote meetings, including the necessary equipment such as a camera for video conferencing, and (b) the use of complex methods used in the research. Such resources and training should be budgeted for in the original funding application

Involvement with different aspects of the research project

The group have enjoyed being involved in so many aspects of the research, from commenting on study materials (e.g., topic guides, questionnaires) to involvement with the public engagement work. We appreciated how researchers genuinely involved us in every area of the project, as well as the positive feedback on our involvement from the research team, which made our hard work and contributions feel appreciated; an example where the group have worked together towards a common purpose and respected different perspectives [14].

When asked to review study materials, our group appreciated the flexibility of the timings to review documents, especially if some members were unable to meet deadlines due to personal circumstances (e.g., caring responsibilities, health reasons). In line with the inclusive opportunities standard of the recently developed framework [14], providing this flexibility meant that all voices were still heard and provided a more inclusive and considerate approach. Being part of this PPIE group allowed members to contribute to research where we felt we were really making a difference, especially those members who have experience of pre-diabetes and diabetes either personally or within their family. It was felt that comments

were taken on board and voices were heard, which felt like 'true PPIE' rather than just a tick box exercise.

Conclusions and recommendations

This commentary has reflected on the PPIE contribution to the DIPLOMA evaluation of the NHS Diabetes Prevention Programme. Based on our learning, we have put together recommendations for other research teams involved in longer-term commissioned research projects like DIPLOMA (see Table 2), in addition to what NIHR already advise as best practice in involving members of the public in research [30]. Many of our reflections correspond with those previously identified in research involving patients and members of the public [6, 16, 31]. However, we have added a unique reflection of PPIE contribution for a longer-term commissioned research evaluation of an NHS service that was already implemented in routine practice, which comes with its own challenges. These reflections and recommendations provide insights to inform future PPIE plans for stakeholders involved in commissioned research evaluations.

Abbreviations

ARC GM Applied Research Collaboration Greater Manchester COVID-19 Coronavirus Disease 2019

DIPLOMA Diabetes Prevention—Long-term Multimethod Assessment

GP General Practitioner

GRIPP2 Guidance for Reporting Involvement of Patients and the Public INVOLVE A national advisory group which promotes and supports greater

public involvement in NHS, public health and social care research

NIHR National Institute for Health and Care Research
NHS National Health Service

PPIE Patient and Public Involvement and Engagement

SSC Study Steering Committee

UK United Kingdom

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s40900-023-00447-0.

Additional file 1. GRIPP2 Short Form Checklist.

Acknowledgements

We would like to extend our thanks to all public contributors, past and present, who have been involved in any aspect of the DIPLOMA research programme, including those involved in the Study Steering Committee Group. We would also like to acknowledge the video producers, scriptwriters and illustrators who have worked with us throughout the DIPLOMA project to create the high quality public engagement materials alongside our PPIE group.

Author contributions

PB, CS and SC secured funding for the DIPLOMA project. PB and CSR had overall responsibility for the PPIE group. MM, EL, HW and CB were all members of the DIPLOMA PPIE group. PB, CS, CSR, SC, KH, LB and REH all had responsibilities for the facilitation of the PPIE group throughout the project duration. REH prepared the manuscript. All authors contributed substantively and read and approved the final manuscript.

Funding

The PPIE group was funded as part of the DIPLOMA research programme. This work is independent research funded by the National Institute for Health and Care Research (The Health and Social Care Delivery Research (HSDR) Programme, 16/48/07—Evaluating the NHS Diabetes Prevention Programme (NHS DPP): the DIPLOMA research programme (Diabetes Prevention—Long Term Multimethod Assessment)). The views and opinions expressed in this manuscript are those of the authors and do not necessarily reflect those of the National Institute for Health and Care Research or the Department of Health and Social Care.

Availability of data and materials

Not applicable.

Declarations

Ethics approval and consent to participate

The DIPLOMA research programme received ethical approval by the North West Greater Manchester East NHS Research Ethics Committee (Reference: 17/NW/0426, 1st August 2017), but ethical approval was not required for the DIPLOMA PPIE group specifically. However, the group provided consent for the research team to securely store some of the personal data (contact details) for the purposes of communication and facilitating reimbursement for their time

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹ Division of Psychology and Mental Health, Faculty of Biology, Medicine and Health, School of Health Sciences, Manchester Centre of Health Psychology, University of Manchester, Manchester, UK. ² Division of Population

Health, Health Services Research & Primary Care, Faculty of Biology, Medicine and Health, NIHR Greater Manchester Patient Safety Translational Research Centre (NIHR PSTRC), Centre for Primary Care and Health Services Research, School of Health Sciences, University of Manchester, Manchester, UK.

³Research & Innovation, Northern Care Alliance NHS Foundation Trust, Salford, UK.

⁴North West Ambulance Service NHS Trust, Bolton, UK.

⁵Division of Population Health, Health Services Research & Primary Care, Faculty of Biology, Medicine and Health, NIHR Applied Research Collaboration Greater Manchester (NIHR ARC GM), Centre for Primary Care and Health Services Research, School of Health Sciences, University of Manchester, Manchester, UK.

⁶Division of Population Health, Health Services Research & Primary Care, Faculty of Biology, Medicine and Health, Centre for Biostatistics, School of Health Sciences, University of Manchester, WA.

⁷DIPLOMA Patient and Public Involvement Group, University of Manchester, Manchester, UK.

⁸DIPLOMA Patient and Public Involvement Group, University of Manchester, Manchester, UK.

Received: 8 March 2023 Accepted: 12 May 2023 Published online: 14 June 2023

References

- Hayes H, Buckland S, Tarpey M. Briefing notes for researchers: public involvement in NHS, public health and social care research. Eastleigh: INVOLVE. 2012. https://www.invo.org.uk/wp-content/uploads/2014/11/ 9938_INVOLVE_Briefing_Notes_WEB.pdf (Accessed 11th August 2022).
- Absolom K, Holch P, Woroncow B, Wright EP, Velikova G. Beyond lip service and box ticking: how effective patient engagement is integral to the development and delivery of patient-reported outcomes. Qual Life Res. 2015;24(5):1077–85.
- Denegri S. Going the extra mile: improving the nation's health and wellbeing through public involvement in research. London: NIHR. 2015. https://www.nihr.ac.uk/documents/about-us/our-contribution-to-research/how-we-involve-patients-carers-and-the-public/Going-the-Extra-Mile.pdf (Accessed 30 January 2023).
- INVOLVE. NIHR Senior Investigators: leaders for patient and public involvement in research. INVOLVE. 2014. https://www.invo.org.uk/wpcontent/uploads/2014/11/9985_INVOLVE_senior_investigators_pub_ WEB.pdf (Accessed 30 January 2023).
- Jenner MK, Gilchrist M, Baker GC. Practical considerations in improving research through public involvement. Res Involv Engag. 2015;1(1):1–6.
- Preston JL, Berryman VR, Hancock A, Pattrick M, Worthington A, Hitman GA, Hood GA. Developing patient and public involvement and engagement (PPIE) in diabetes research: a local approach. Pract Diabetes. 2019;36(3):81–5.
- NHS England, Care Quality Commission, Health Education England, Monitor, Public Health England, Trust Development Authority. Five Year Forward View. 2014. http://www.england.nhs.uk/wp-content/uploads/ 2014/10/5vfv-web.pdf (Accessed 30 January 2023).
- Diabetes statistics: diabetes prevalence 2019. Diabetes UK. 2020. https:// www.diabetes.org.uk/professionals/position-statements-reports/stati stics/diabetes-prevalence-2019 (Accessed 30 January 2023).
- Sood HS, Maruthappu M, Valabhji J. The National Diabetes Prevention Programme: a pathway for prevention and wellbeing. Br J Gen Pract. 2015;65(636):336–7.
- NHS England Diabetes Prevention Programme (NHS DPP). NHS (2021). https://www.england.nhs.uk/diabetes/diabetes-prevention/ (Accessed 30 January 2023).
- Valabhji J, Barron E, Bradley D, Bakhai C, Fagg J, O'Neill S, Young B, Wareham N, Khunti K, Jebb S, Smith J. Early outcomes from the English National health service diabetes prevention programme. Diabetes Care. 2020;43(1):152–60.
- Sutton M. Evaluating the NHS Diabetes Prevention Programme (NHS DPP): the DIPLOMA research programme (Diabetes Prevention Long term Multimethod Assessment). Health Services and Delivery Research programme, project. 2017; https://www.journalslibrary.nihr.ac.uk/programmes/hsdr/164807/#/ (Accessed 30 January 2023).
- Staniszewska S, Brett J, Simera I, Seers K, Mockford C, Goodlad S, Altman DG, Moher D, Barber R, Denegri S, Entwistle A. GRIPP2 reporting checklists: tools to improve reporting of patient and public involvement in research. BMJ. 2017;358.

- UK Public Involvement Standards Development Partnership: Better public involvement for better health and social care research. UK Standards for Public Involvement. NIHR. 2019; 12. https://sites.google.com/nihr.ac.uk/ pi-standards/standards (Accessed 30 January 2023).
- NIHR. Research Governance Guidelines. 2022; Version 3, January 2023. https://www.nihr.ac.uk/documents/research-governance-guidelines/ 12154 (Accessed 01 March 2023).
- Harrison JD, Auerbach AD, Anderson W, Fagan M, Carnie M, Hanson C, Banta J, Symczak G, Robinson E, Schnipper J, Wong C. Patient stakeholder engagement in research: a narrative review to describe foundational principles and best practice activities. Health Expect. 2019;22(3):307–16.
- NIHR. Payment guidance for researchers and professionals. NIHR Resources. 2022;Version 1.3, July 2022. https://www.nihr.ac.uk/documents/payment-guidance-for-researchers-and-professionals/27392 (Accessed 30 January 2023).
- NIHR Applied Research Collaboration Greater Manchester. DIPLOMA Project. [Published 20 July 2018]. https://www.youtube.com/watch?v= W8ql4zQ2F9E (Accessed 30 January 2023).
- NIHR Applied Research Collaboration Greater Manchester. DIPLOMA: Findings from the evaluation of Healthier You: The NHS Diabetes Prevention Programme. [Published 16 June 2022]. https://www.youtube.com/watch?v=7vJdeUCPiil (Accessed 30 January 2023).
- NIHR Applied Research Collaboration Greater Manchester. DIPLOMA: Behaviour Change. [Published 22 May 2023]. https://www.youtube.com/watch?v=mmKMr_7RNaM (Accessed 22 May 2023).
- NIHR Applied Research Collaboration Greater Manchester. Taking steps towards preventing diabetes: Our findings on access to the NHS Healthier You Programme. [Published 14 November 2021]. https://www.youtube.com/ watch?v=68uzBVX9F8I (Accessed 30 January 2023).
- NIHR Applied Research Collaboration Greater Manchester. Preventing type 2 diabetes by going digital: What's the evidence? Findings from the DIPLOMA research programme. [Published 21 March 2023]. https://www.youtube. com/watch?v=T2JVWAb6Ptw (Accessed 22 March 2023).
- Howells K, Bower P, Burch P, Cotterill S, Sanders C. On the borderline of diabetes: understanding how individuals resist and reframe diabetes risk. Health Risk Soc. 2021;23(1–2):34–51.
- Ross JA, Cotterill S, Bower P, Murray E. Influences on patient uptake of and engagement with the national health service digital diabetes prevention programme: qualitative interview study. J Med Int Res. 2023;25: e40961.
- Smits DW, Van Meeteren K, Klem M, Alsem M, Ketelaar M. Designing a tool to support patient and public involvement in research projects: the Involvement Matrix. Res Involv Engag. 2020;6(1):1–7.
- NIHR Applied Research Collaboration Greater Manchester. What is the Diabetes Prevention Programme and why is it important? [Published 13 June 2018]. https://arc-gm.nihr.ac.uk/news/blog/Blog-Carole-Bennett-DIPLOMA-PPI (Accessed 30 January 2023).
- NIHR Applied Research Collaboration Greater Manchester. Eric Lowndes talks about his experiences of attending an NHS behaviour change programme and working with two independently funded evaluation teams. [Published November 2020]. https://arc-gm.nihr.ac.uk/news/blog/ BLOG-Eric-Lowndes-PPI-evaluations-of-national-behaviour-change-programmes (Accessed 30 January 2023).
- 28. Adeyemi I, Sanders C, Ong BN, Howells K, Quinlivan L, Gorman L, Giles S, Amp M, Monaghan E, Naseem S, Pearson A. Challenges and adaptations to public involvement with marginalised groups during the COVID-19 pandemic: commentary with illustrative case studies in the context of patient safety research. Res Involv Engag. 2022;8(1):1–2.
- 29. Russell J, Fudge N, Greenhalgh T. The impact of public involvement in health research: What are we measuring? Why are we measuring it? Should we stop measuring it? Res Involv Engag. 2020;6(1):1–8.
- NIHR. Briefing notes for researchers—public involvement in NHS, health and social care research. NIHR Resources. 2021;Version 1.0, April 2021. https://www.nihr.ac.uk/documents/briefing-notes-for-researchers-public-involvement-in-nhs-health-and-social-care-research/27371 (Accessed 30 January 2023).
- Brett JO, Staniszewska S, Mockford C, Herron-Marx S, Hughes J, Tysall C, Suleman R. A systematic review of the impact of patient and public involvement on service users, researchers and communities. Patient-Patient-Centered Outcomes Res. 2014;7(4):387–95.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

